Author’s response to reviews

Title: Effect of a lifestyle-focused electronic patient support application for improving risk factor management, self-rated health and prognosis in post myocardial infarction patients: a study protocol for a multi-center randomized controlled trial

Authors:

Manuel Gonzalez Garcia (manuel.gonzalez-garcia@csiro.au)

Ingela Sjölin (ingela.sjolin@gmail.com)

Maria Bäck (maria.back@liu.se)

Halldora Ögmundsdottir Michelsen (halldora.michelsen@med.lu.se)

Tina Tanha (tina.tht@gmail.com)

Camilla Sandberg (camilla.sandberg@umu.se)

Alexandru Schiopu (alexandru.schiopu@med.lu.se)

Margret Leosdottir (margret.leosdottir@med.lu.se)

Version: 1 Date: 19 Oct 2018

Author’s response to reviews:

Dear Editor,

We have revised the manuscript TRLS-D-18-00055, titled “Effect of a lifestyle-focused electronic patient support application for improving risk factor management, self-rated health and prognosis in post myocardial infarction patients: a study protocol for a multi-center randomized controlled trial”, according to comments raised by the reviewers. Attached you will find point-to-point responses to the comments, the reviewed manuscript with comments from the authors (ML) and the final manuscript as a Word doc.

We hope that you will consider the manuscript for publication.

Sincerely,